Author's response to reviews

Title: Oral administration of PPC enhances antigen-specific CD8+ T cell responses while reducing IgE levels in sensitized mice.

Authors:

Mike Burrows (mburrows@tampabayresearch.org)
Deepak Assundai (Deepak.Assundai@moffitt.org)
Esteban Celis (Esteban.Celis@moffitt.org)
Frank Tufaro (ftufaro@mac.com)
Akiko Tanaka (atanaka@tampabayresearch.org)
William G Bradley (gbradley@tampabayresearch.org)

Version: 4 Date: 11 November 2009

Author's response to reviews: see over
November 11, 2009

To the Editor of BMC Complementary and Alternative Medicine:

Re: Manuscript 9973033993017222; Oral administration of PPC enhances antigen-specific CD8+ T cell responses while reducing IgE levels in sensitized mice, authored by Burrows et al.

We have responded to the request to add a Statistical Analysis description in the Methods section and have uploaded the corrected manuscript.

Respectfully,

W. Guy Bradley, Ph.D.

Tampa Bay Research Institute
10900 Roosevelt Blvd N
St. Petersburg, FL 33716
Telephone: 727.576.6675
Fax: 727.577.9892
E-mail: gbradley@tampabayresearch.org